Author's response to reviews

Title: Increased expression of vascular endothelin type B and angiotensin type 1 receptors in patients with ischemic heart disease

Authors:

Ivan Dimitrijevic (ivan.dimitrijevic@med.lu.se)
Marie-Louise Edvinsson Edvinsson (MarieLouise.L.Edvinsson@skane.se)
Qingwen Chen (cherqw@hotmail.com)
Malin Malmsjö (malin.malmsjo@med.lu.se)
Per-Ola Kimblad (Per-Ola.Kimblad@skane.se)
Lars Edvinsson (lars.edvinsson@med.lu.se)

Version: 7 Date: 23 July 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

Tables must be created using a Tables tool

Reply; Table 1 is now corrected and created using the Table tool.

Sincerely

Ivan Dimitrijevic, MD